Author's response to reviews

Title: Mediastinal hydatid cyst: A case report

Authors:

msougar yassine (msougar@hotmail.com)
Oussama Afandi (dr.afandioussama@gmail.com)
Nadia Ihfa (nadia.ihfa@yahoo.fr)
Youness Rouiessi (younnessrouiessi@gmail.com)
Mohamed Khallouki (mohamedkhallouki@hotmail.com)
Benacer Finech (benacer.finech@yahoo.fr)

Version: 2 Date: 31 May 2013

Author's response to reviews:

We try to answer to reviewers recommendatons.
First we delete all abreviation from the abstract, reform the time using for the text, and make some language corrections.
According to reviewers comments, we reform the whole case presentation, add ethnical background of the patient and including a conclusion wich was forgotten in the first manuscript.
Thanks